Understanding the health-care experiences of people with sickle cell disorder transitioning from paediatric to adult services: This Sickle Cell Life, a longitudinal qualitative study

Alicia Renedo,¹ Sam Miles,¹ Subarna Chakravorty,² Andrea Leigh,³ John O Warner^{4,5} and Cicely Marston^{1*}

¹Department of Public Health, Environments and Society, Faculty of Public Health and Policy, London School of Hygiene & Tropical Medicine, London, UK ²Department of Paediatric Haematology, King's College Hospital, London, UK ³University College London NHS Hospitals Foundation Trust, London, UK ⁴National Heart and Lung Institute, Imperial College London, London, UK ⁵Collaboration for Leadership in Applied Health Research and Care for Northwest London, Imperial College London, London, UK

*Corresponding author Cicely.Marston@lshtm.ac.uk

Declared competing interests of authors: Subarna Chakravorty reports receiving honoraria for taking part in advisory boards for Novartis Pharmaceuticals UK Ltd (London, UK) and an honorarium from Novartis Pharmaceuticals UK Ltd for participating in the development of a national consensus guideline document for the management of health-care transition services in sickle cell disorder. John O Warner reports acting as a principal investigator and a scientific advisory board member (2005 to present) for trials of infant milk formulae for the prevention of allergy (Danone SA, Paris, France/Nutricia, Zoetermeer, the Netherlands); acting as an investigator and a scientific advisory board member (2008 to present) for trials of an environmental control system for allergic conditions (Airsonett AB, Ängelholm, Sweden); bursaries for lectures at conferences for Danone SA/Nutricia, Airsonett AB, UCB Pharma (Brussels, Belgium), Novartis Pharmaceuticals UK Ltd and Allergy Therapeutics plc (Worthing, UK); acting as medical advisor (1993 to present) for the Anaphylaxis Campaign (www.anaphylaxis.org.uk); membership of the scientific advisory board (2017 to present) for a UK government Home Office enquiry into potential toxic health hazards from the Grenfell Tower fire; and acting as academic theme lead (2014–19) for the Collaboration for Leadership in Applied Health Research and Care for Northwest London (until 30 September 2019).

Published November 2020 DOI: 10.3310/hsdr08440

Plain English summary

This Sickle Cell Life, a longitudinal qualitative study Health Services and Delivery Research 2020; Vol. 8: No. 44 DOI: 10.3310/hsdr08440

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

What happens during transitions from childhood to adulthood can affect young people's future lives. Children with long-term health conditions must learn to manage their own health as they get older and eventually start using adult health-care services. These transitions to adult care do not always go well. Sickle cell disorder is one long-term condition for which transitions are particularly challenging and people's health can suffer.

We investigated the experiences of young people with sickle cell disorder as they transitioned to adult services. We explored what else might affect their health-care transitions outside the hospital. We interviewed young people with sickle cell disorder and sickle cell disease specialist health-care providers in two cities in England.

Key obstacles young people face during transition include not being listened to by others (e.g. teachers, friends, non-specialist hospital staff) and not being recognised as experts in their own condition. Healthcare providers and other adults want young people to learn about their condition and act responsibly to stay healthy. However, when young people try to take responsibility (e.g. making requests about the care they receive) they are often ignored. As a result, young people stop trusting places where they have not experienced good care, such as in accident and emergency departments and on general wards, and avoid going, although they know that avoiding hospital is risky.

Such young people also struggle to develop adult identities (i.e. ideas about how they should behave as adults). They want to work hard and do well in school, but this is incompatible with their need to take responsibility for their condition, for instance needing to rest to avoid pain. These conflicting demands are hard to balance.

We need to support others in young people's social and health-care environments to learn about sickle cell disorder and to treat young people as experts in their own condition.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Renedo *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/ health-services-and-delivery-research.htm

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 13/54/25. The contractual start date was in April 2015. The final report began editorial review in December 2019 and was accepted for publication in June 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Renedo *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of Health Services and Delivery Research and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk